Author's response to reviews

Title: Relationships among neurocognition, symptoms and functioning in patients with schizophrenia: a path-analytic approach for associations at baseline and following 24 seeks of antipsychotic drug therapy

Authors:

Ilya Lipkovich (lipkovichia@lilly.com)
Deberdt Walter (deberdt_walter@lilly.com)
John G Csernansky (jgc@conte.wustl.edu)
Bernard Sabbe (bernard.sabbe@ua.ac.be)
Richard SE Keefe (richard.keefe@duke.edu)
Sara Kollack-Walker (Kollack-Walker_Sara@lilly.com)

Version: 3 Date: 12 July 2009

Author's response to reviews: see over
July 9, 2009

Dear Editor:

We have completed the required revisions to our manuscript (MS: 1466203774226218) focusing on neurocognition, symptoms and functioning following antipsychotic drug therapy. The manuscript is titled “Relationships Among Neurocognition, Symptoms and Functioning in Patients with Schizophrenia: A path-Analytic Approach for Associations at Baseline and Following 24 Weeks of Antipsychotic Drug Therapy.” We would like to have our revised manuscript considered for publication in *BMC Psychiatry*. We have also attached a detailed response to the latest set of comments.

As noted previously, the data presented in this manuscript have not been published previously and are not under consideration for publication elsewhere. All authors have made a substantive contribution to the work, and each has read and approved the manuscript.

Please address all correspondence to either: Dr. Ilya Lipkovich, Ph.D., Lilly Research Laboratories, Eli Lilly and Company, Lilly Corporate Center, DC 6054, Indianapolis, IN 46285; Telephone: (317) 651-6095; Facsimile: (317) 433-1304; E-mail: lipkovichia@lilly.com, or Dr. Sara Kollack-Walker, Ph.D., Lilly Research Laboratories, Eli Lilly and Company, Lilly Corporate Center, Indianapolis, IN 46285; Telephone: (317) 433-4654; Facsimile: (317) 276-7100; Kollack-Walker_Sara@lilly.com.

Sincerely,

Ilya Lipkovich, Ph.D.  
Sara Kollack-Walker, Ph.D.
Dear Dr Kollack-Walker,

We are delighted to confirm that BMC Psychiatry will publish your manuscript (above) pending formatting requirements. We ask that you make the following changes to the format of your manuscript so that online production process of your manuscript is straightforward. After making these changes, please resubmit a new version of your paper to the submission system. The instructions to submit your revised manuscript are available below.

Please ensure that all revisions are made to the most recent version of your article which can be accessed at the following link:
http://www.biomedcentral.com/media/man/1466203774226218/msd/main/0

Figures:
http://www.biomedcentral.com/media/man/1466203774226218/msd/FIG/1
http://www.biomedcentral.com/media/man/1466203774226218/msd/FIG/2
http://www.biomedcentral.com/media/man/1466203774226218/msd/FIG/3
http://www.biomedcentral.com/media/man/1466203774226218/msd/FIG/4

Formatting changes requested
-----------------------------

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. REPLY: This has been checked and confirmed to match.

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. REPLY: We have reviewed the addresses and made corrections where necessary.

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Authors; information (optional); Acknowledgements (optional); References; Figure legends (if any); Tables (if any); Description of Additional files (if any). REPLY: We have reviewed the organization of the manuscript, removing the Limitations subheading and confirming that the sections retained follow this order.
Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Study¿. To `The title of my prestigious publication: a scientific study;Reply: We have made this change.

'HGGN Cognition
6/30/2009'
Please remove this text from the head of each page. Reply: We have made this change.

Keywords, Word-counts and Running title ¿ Please remove these. Reply: We have made these changes.

'Email addresses'
Please remove this subheading from the title page. Reply: We have made this change.

Tracking - Please remove all tracking from the manuscript. Reply: We have made this change.

Abstract - Please ensure that the Abstract is present in manuscript and matches exactly what is in the submission system. Reply: We have checked and can confirm that the abstract is present in the manuscript and matches exactly to what is in the submission system.

Tables - Please do not include vertical lines in your tables. Reply: We have removed vertical lines from the tables.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures Reply: We have reviewed the guide regarding the cropping of figures. We have placed the figures in .tif format.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance. Reply: We have reviewed the manuscript again for any typographical errors.
Thank you
-----------------------------

Please note that this is the proofing stage. All revised manuscripts submitted to BMC Psychiatry are regarded as being in the final version. You will not be able to make any corrections to your article once it has been accepted and published in preliminary form. Therefore, when you submit your final version,
please carefully check your manuscript for any typing or content errors that might have inadvertently occurred, and for small changes to the text that you or your co-authors wish to make at this time. This will be the last chance you have to make any such corrections as any errors present in the manuscript may end up in the final published version. Please submit the revised version of your manuscript via the online submission system:
http://www.biomedcentral.com/author/manuscript/details/view.do?manuscriptId=1466203774226218

To submit your revised manuscript
----------------------------------

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.
2. Click on the 'Submit revision' button next to the title of this manuscript to go through to a submission form.
3. With the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version. It is very important that all changes are updated on this page, as well as in the manuscript file as the information on this page will be used in PubMed and on BioMed Central.
4. With the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.
5. With the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments. Please do not hesitate to contact us if you have any problems or questions.

We would ask that you return any revisions by Tuesday, 14 July 2009.

With all best wishes,

The BioMed Central Editorial Production Team

Tel: +44 (0) 20 3192 2014
Fax: +44 (0) 20 3192 2012
e-mail: corepreaccept@biomedcentral.com
Web: http://www.biomedcentral.com/